Author's response to reviews

Title: Systematic Review: A Single Item Measure of Self-Rated Mental Health

Authors:

Farah Ahmad (farahmad@yorku.ca)
Anuroop K Jhajj (anu333@gmail.com)
Donna E Stewart (donna.stewart@uhn.ca)
Madeline Burghardt (mb0612rm@yorku.ca)
Arlene S Bierman (arlene.bierman@utoronto.ca)

Version: 2 Date: 18 March 2014

Author's response to reviews:

Following changes have been made:

ABSTRACT
1. The revised abstract now has following sections:
   - Background
   - Methods
   - Results
   - Conclusions
2. Several abbreviations have been removed.
3. The word count is 323

RESEARCH ARTICLE
The revised version now has following sections:
   - Background
   - Methods
   - Results
   - Discussion
   - Conclusion

LINE NUMBERING
Line numbering has been added